Author's response to reviews

Title: Seasonal variation in orthopedic health services utilization in Switzerland: The impact of winter sport tourism.

Authors:

Klazien Matter-Walstra (klazien.matter@memcenter.unibe.ch)
Marcel Widmer (marcel.widmer@memcenter.unibe.ch)
Andre Busato (andre.busato@memcenter.unibe.ch)

Version: 4  Date: 21 February 2006

Author's response to reviews:

Title and Abstract: identical in submission system and manuscript
Authorist: identical in submission system and manuscript
Title: Full stop removed
Competing interests changed in:
The authors declare that they have no competing interests. This project was supported by the National Research Program NRP 53 "Musculoskeletal Health - Chronic Pain" of the Swiss National Science Foundation (Project 405340-104607)
Table: gray shading removed and changed into bold table border